Author's response to reviews

Title: Central coordination as an alternative for local coordination in a multicenter randomized controlled trial: the FAITH trial experience

Authors:

Stephanie M Zielinski (s.zielinski@erasmusmc.nl)
Helena Viveiros (viveiro@mcmaster.ca)
Martin J Heetveld (heetveld@kg.nl)
Marc F Swionkowski (swion001@umn.edu)
Mohit Bhandari (bhandam@mcmaster.ca)
Peter Patka (p.patka@erasmusmc.nl)
Esther MM Van Lieshout (e.vanlieshout@erasmusmc.nl)

Version: 3 Date: 28 December 2011

Author's response to reviews: see over
To the Editor-in-Chief of Trials

Dear Prof. Furberg,

We hereby respectfully resubmit the enclosed revised manuscript by S.M. Zielinski, E.M.M. Van Lieshout, H. Viveiros, M.J. Heetveld, M.F. Swiontkowski, M. Bhandari, P. Patka, on behalf of the FAITH trial investigators, entitled “Central coordination as an alternative for local coordination in a multicenter randomized controlled trial: the FAITH trial experience” for publication in Trials.

All the revisions concern the formatting changes that were requested by the editorial production team. We hope that we have provided satisfactory changes to our manuscript.

Sincerely,

Stephanie M. Zielinski, MD
(First author)

Esther M.M. Van Lieshout, PhD
(Corresponding author)